Author's response to reviews

Title: Mental health (GHQ12; CES-D) and attitudes towards the value of work among inmates of a semi-open prison and the long-term unemployed in Luxembourg

Authors:

Michele Baumann (michele.baumann@uni.lu)
Raymond Meyers (raymond.meyers@uni.lu)
Etienne Le Bihan (etienne.lebihan@uni.lu)
Claude Houssemand (claude.houssemand@uni.lu)

Version: 6 Date: 10 June 2008

Author's response to reviews: see over
Monday, June 10, 2008

Dear Dr Bucceri,

RE: urgent Formatting Changes for MS: 2034862742168012

Please find enclosed the revised version of the above manuscript according to the asked changes.

Looking forward to hearing from the Editorial Board.

Sincerely yours

Prof. Michèle Baumann